Author's response to reviews

Title: Malignant transformation of oral potentially malignant disorders: a retrospective cohort study

Authors:

Pei-Shan Ho (psho@kmu.edu.tw)
Yi-Hsin Yang (yihsya@kmu.edu.tw)
Pai-Li Chen (scalpel58@seed.net.tw)
Saman Warnakulasuriya (s.warne@kcl.ac.uk)
Tien-Yu Shieh (tiyush@kmu.edu.tw)
Yun-Kwan Chen (yukkwa@kmu.edu.tw)
Chung-Ho Chen (chench@kmu.edu.tw)
I-Yueh Huang (iyuhu@kmu.edu.tw)

Version: 3 Date: 31 October 2008

Author's response to reviews:

1. In the previous submission, we forgot to add a cover letter. Now we try to correct it and resubmit again.
2. We have formatted my abstract according to the instructions for authors.
3. We added a section in my methods section stating relevant ethical approval for my works.